Author's response to reviews

Title: Associations between cadmium exposure and neurocognitive test scores in a cross-sectional study of US adults

Authors:

Timothy Ciesielski (timothyhciesielski@gmail.com)
David C Bellinger (david.bellenger@childrens.harvard.edu)
Joel Schwartz (jschwtz@hsph.harvard.edu)
Russ Hauser (rhauser@hohp.harvard.edu)
Robert O Wright (robert.wright@mssm.edu)

Version: 3 Date: 29 January 2013

Author's response to reviews: see over
Dear EH Editorial Staff,

This is the reformatted version of the manuscript: *Associations between cadmium exposure and neurocognitive test scores in a cross-sectional study of US adults*. The tables are now in MS word table format, the reference list changes have been made. We provide detailed point by point responses to the formatting requests on the following page below.

Please let me know if you find any additional issues that need to be addressed. Thank you very much to the editors and reviewers for your fast response to our resubmission, and for your help in improving this manuscript.

Sincerely,

Timothy Ciesielski
As for the formatting, the numbers before the institutional addresses on the title page should be superscript.
Done.

Remove the wording (Corresponding author) and insert the phrase *Corresponding author above the email addresses. Also place the * after the corresponding author's superscript number(s).
Done.

Please format the Email addresses as author's initials:email address e.g. JS:joe.shmoe@university.edu.
Done.

Change all underlined text to normal i.e. no underline.
Done.

In the Abstract remove the colons and place the text below the headings.
Done.

Remove the line numbering.
Done.

The List of abbreviations should be formatted as abbreviation:term and separating the pairs with semi-colons in sentence format.
Done.

In the References remove the issue numbers and list up to 30 authors before using et al.
Done.

For the tables, all horizontal lines should be visible.
Done.

Please also ensure that your revised manuscript conforms to the journal style (http://www.ehjournal.net/info/instructions/). It is important that your files are correctly formatted.
It looks like the primary remaining issue was getting the tables into MS Word table object format to preserve the proper alignment of columns, and this has been done.

Please let me know if you find any additional issues that need fixing and thank you very much for your help in improving this manuscript.